Author's response to reviews

Title: Palliative care in advanced dementia; a mixed methods approach for the development of a complex intervention

Authors:

Elizabeth L Sampson (e.sampson@medsch.ucl.ac.uk)
Ingela Thuné-Boyle (i.thune-boyle@medsch.ucl.ac.uk)
Riitta Kukkastenvehmas (r.kukkastenvehmas@medsch.ucl.ac.uk)
Louise Jones (GORDJOCL@aol.com)
Adrian Tookman (adrian.tookman@royalfree.nhs.uk)
Michael King (m.king@medsch.ucl.ac.uk)
Martin R Blanchard (m.blanchard@medsch.ucl.ac.uk)

Version: 3 Date: 16 June 2008

Author's response to reviews:

Dear Editor,

We have made the formatting changes as requested:

1. Each affiliation has been written in full, in the following format: department, institute, city, and country.
2. I have checked and removed full-stops in abbreviated country names
3. I have linked each author to their corresponding affiliation with the use of a superscript number

With Best Wishes
Elizabeth Sampson
Corresponding Author